Dear Editor,

We would like to thank you for the attention given to our paper entitled “Assessing Moroccan Physician Knowledge and Practices Regarding Maternal Obesity’s Impact on Childhood Obesity: Implications for Prevention and Intervention” and for your help, which has been very useful to improve our manuscript. Please find below the answers to your and the referee’s queries.

Best regards,

DOUIYEH Imane, corresponding author

➢ **Introduction:** please add a definition of obesity in childhood.

Thank you for your suggestion. We have taken your recommendation into account and added a detailed definition of childhood obesity in the Introduction section for a better understanding of the context of our study. We hope this change meets your expectations.

➢ **Material and methods:** number of approval by ethical committee should be added.

Thank you for your suggestion. However, it is important to note that our study did not involve human or animal samples. We surveyed physicians who were members of the association with which we collaborated on our study. Therefore, no specific ethical approval was required for our methodology. For greater clarity, please find attached the document attesting to our collaboration with the medical association.

➢ **Results:** the important difference between male and female should be better explained.

We have taken your suggestion on board and added a more in-depth analysis to explain more clearly the significant difference between men and women in the results.

➢ **Discussion:** is very long and repetitive. Can be improved.

We have taken your suggestions into consideration and made significant changes to the discussion. I’ve reduced its length by eliminating repetition and maintaining conciseness while preserving the integrity of key points. I hope these adjustments meet your expectations.